Response to Reviewer

thanks for the reviewing the manuscript.

Reviewer’s Question/comments:

At present, the epidemiological characteristics of urinary system injury in sub Saharan region are relatively blank. Through literature review and analysis, the author collected data as much as possible and summarized the epidemiology and characteristics of urinary system injury in this region, which is very helpful for us to understand the characteristics of urinary system injury in this area. Although the situation in this region is complex, and some regional differences may be large, the overall data is of reference significance, and it is suggested that the language should be revised for publication.

Response: Many thanks for the reviewing the manuscript and providing insight. The English has been revised as recommended

EDITORIAL OFFICE’S COMMENTS:

The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor; and (2) PMID and DOI numbers are missing in the reference list. Please provide the PubMed numbers and DOI citation numbers to the reference list and list all authors of the references. Please revise throughout.

Response:

Many thanks for the editorial work.

(1) The original figure document has been provided in an editable format.  
(2) PMID and DOI numbers have been placed along the respective references as provide by the journal. However, a few citations do not have doi numbers assigned by their journals.  
(3) All authors have been listed in each reference/citation.